University of Texas Rio Grande Valley

ScholarWorks @ UTRGV

Economics and Finance Faculty Publications and Presentations

Robert C. Vackar College of Business & Entrepreneurship

4-2021

Organizational Resources, Country Institutions, and National Culture behind Firm Survival and Growth during COVID-19

Yu Liu

The University of Texas Rio Grande Valley, yu.liu@utrgv.edu

Mike W. Peng

Zuobao Wei

Jian Xu

The University of Texas Rio Grande Valley, jian.xu01@utrgv.edu

Lixin Colin Xu

Follow this and additional works at: https://scholarworks.utrgv.edu/ef_fac



Part of the Finance Commons

Recommended Citation

Liu, Yu; Peng, Mike W.; Wei, Zuobao; Xu, Jian; Xu, Lixin Colin. 2021. Organizational Resources, Country Institutions, and National Culture behind Firm Survival and Growth during COVID-19. Policy Research Working Paper; No. 9633. World Bank, Washington, DC.

This Article is brought to you for free and open access by the Robert C. Vackar College of Business & Entrepreneurship at ScholarWorks @ UTRGV. It has been accepted for inclusion in Economics and Finance Faculty Publications and Presentations by an authorized administrator of ScholarWorks @ UTRGV. For more information, please contact justin.white@utrgv.edu, william.flores01@utrgv.edu.

Policy Research Working Paper

9633

Organizational Resources, Country Institutions, and National Culture behind Firm Survival and Growth during COVID-19

Yu Liu
Mike W. Peng
Zuobao Wei
Jian Xu
Lixin Colin Xu



Abstract

This paper provides one of the first comprehensive and most updated studies on the effects of firms' organizational resources, country institutions, and national culture on the survival and growth of private firms around the world during the COVID-19 pandemic. Analyzing World Bank Enterprise Follow-up Surveys on COVID-19 that cover 18,770 firms in 36 countries, the paper documents four sets of findings. (1) During the pandemic, firms with favorable organizational resources (state ownership and affiliation with parent companies) are more likely to survive and grow, whereas firms with foreign ownership or more financial obstacles are less likely to survive or grow. Firms in countries with a higher per capita income, a lower COVID-19

spread, and a less stringent COVID-19 control policy are more likely to survive and grow. (2) Favorable ownership and parent-company affiliations help cushion the pandemic shock during the pandemic. (3) The relationship between firm characteristics and firm survival/growth is significantly affected by the stringency of a country's COVID-19 policy. (4) Firm survival and growth are positively related to a country's cultural tendency in terms of long-term orientation and are not significantly related to uncertainty avoidance and individualism. The overall quality of country governance is negatively linked to the odds for firm survival as well as revenue and employment growth.

This paper is a product of the Development Research Group, Development Economics. It is part of a larger effort by the World Bank to provide open access to its research and make a contribution to development policy discussions around the world. Policy Research Working Papers are also posted on the Web at http://www.worldbank.org/prwp. The authors may be contacted at lxu1@worldbank.org.

The Policy Research Working Paper Series disseminates the findings of work in progress to encourage the exchange of ideas about development issues. An objective of the series is to get the findings out quickly, even if the presentations are less than fully polished. The papers carry the names of the authors and should be cited accordingly. The findings, interpretations, and conclusions expressed in this paper are entirely those of the authors. They do not necessarily represent the views of the International Bank for Reconstruction and Development/World Bank and its affiliated organizations, or those of the Executive Directors of the World Bank or the governments they represent.

Organizational Resources, Country Institutions, and National Culture behind Firm Survival and Growth during COVID-19¹

Yu Liu

University of Texas Rio Grande Valley Edinburg, TX 78539 Email: yu.liu@utrgv.edu

Mike W. Peng

University of Texas at Dallas Richardson, TX 75080 Email: mikepeng@utdallas.edu

Zuobao Wei

University of Texas at El Paso El Paso, TX 79968 Email: zwei@utep.edu

Jian Xu

University of Texas Rio Grande Valley Edinburg, TX 78539 Email: jian.xu01@utrgv.edu

Lixin Colin Xu

The World Bank Washington, DC 20433 Email: lxu1@worldbank.org

JEL classifications: G01, G32, G38

Keywords: COVID-19, private firms, firm survival and growth, organizational resources

¹ The paper reflect the authors' view and do not implicate the World Bank and its member countries.

Bad companies are destroyed by crisis. Good companies survive them. Great companies are improved by them.

— Andy Grove, Former Intel Corporation CEO

1. Introduction

The COVID-19 outbreak, starting in late 2019, was declared by the World Health Organization as a global pandemic on March 11, 2020. Since then, the virus has quickly spread around the world and claimed more than 2.6 million lives worldwide as of March 31, 2021 (www.who.int). While the pandemic has extracted huge loss of human lives, the loss in economic terms is also tremendous. Hence, the once-in-a-life time COVID-19 shock has attracted unprecedented attention from scholars of all disciplines, and deservedly so.

Scholars have acted quickly to point out that the pandemic has adversely affected firm operating performance and stock returns around the world (Alfaro et al. 2020; Al-Awadhi et al. 2020; Ashraf 2020b; Ramelli and Wagner 2020). While this emerging literature on corporate immunity has paid most attention to listed firms in the United States,² some studies have examined the effects of firm/country characteristics on listed firms in a cross-country setting (Ding et al. 2021). In a cross-country study of private firms in 13 countries, Chundakkadan, Raj, and Sasidharan (2020) focus on how financial constraint impacts firm survival rate, and Liu, Wei, and Xu (2021) examine the relative impact of the pandemic on female-headed firms in 24 countries. In single-country studies, Carletti et al. (2020) and Gu et al. (2020) examine Italian and Chinese private firms' behaviors and outcomes during the pandemic, respectively. Despite this emerging literature, on a *worldwide* basis, much remains unknown: How are firm survival and growth affected by firms' access to resources around the world? Do a country's institutions and culture affect firm survival and growth during the pandemic? If so, how? How do firms behave differently during normal times and during the pandemic times?

The pandemic offers a unique context to enhance our understanding of interactions between institutions and organizations. How institutions and organizations interact with each other is a key theme for research in major social sciences disciplines, such as economics (North, 1990), management (Peng, 2003), and sociology (Scott, 2014). A consistent theme is that firms on the one hand leverage their organizational resources, and on the other hand are enabled and constrained by the formal and informal institutional frameworks in which they are embedded. Given the magnitude of this pandemic crisis, firms are likely to endeavor to do their best—

² Albuquerque et al. (2020); Alfaro et al. (2020); Bloom et al. (2020, 2021); Bae et al. (2021); Demers et al. (2020); Fahlenbrach, Rageth and Stulz (2020); Li et al. (2020); Pagano, Wagner and Zechner (2020); Papanikolaou and Schmidt (2020); and Ramelli and Wagner (2020).

survival is at stake. As a result, exploring the impact of organizational resources, national institutions, and culture on the survival and growth of firms is not only of significant scholarly interest, but is also of obvious importance to practitioner and policy interest.

In this study, we add to the burgeoning literature on COVID-19 and firms around the world by conducting one of the first comprehensive cross-country studies concerning the impact of firm/country characteristics on the survival and growth of private firms in mostly developing countries during the pandemic. Our paper builds on and yet differs from the aforementioned studies in at least three aspects. First, we focus on private (unlisted) firms in mostly developing countries, whereas most of the aforementioned COVID studies focus on listed firms. The private sector plays an increasingly important role in developing economies around the world. For the developing world, the private sector provided 90% of employment, 60% of investment, and more than 80% of government revenues.³ From the onset of the COVID-19 crisis, small and medium-sized enterprises (SMEs), which constitute an overwhelming majority of our sample firms, have suffered tremendous operational and financial interruptions—more so than the listed firms. According to World Bank Enterprise Follow-up Surveys on COVID-19 (WBES-COVID19), which have been completed in 36 countries as of February 2021, 11% of surveyed firms are temporarily or permanently closed. The closure rate is 8.1% for listed firms and 11.6% for unlisted/private firms (the rate for SMEs is even higher at 11.8%). Since business obstacles tend to have a more pronounced impact on small firms than on large firms (Beck, Demirgüç-Kunt, and Maksimovic, 2005; Knack and Xu, 2017), it is reasonable to conjecture that the pandemic shock has a more pronounced impact on private firms than on listed firms.

Second, compared to single-country studies (Carletti et al., 2020; Dai et al., 2021a, 2021b; Gu et al., 2020;), our cross-country setting allows us to examine the effects of cross-country variations in COVID spread severity, COVID policy stringency, country governance, and culture on private firms' survival and growth. We thus cover more comprehensively various key firm/country aspects that could affect the pandemic impact.

Third, our study covers a significantly longer time period into the pandemic than the aforementioned studies. The above-cited cross-country firm-level studies use a relatively short time period, the majority of which examines a period from January to March 2020, with a few

³ The source of these statistics is the Australian Department of Foreign Affairs and Trade.

https://www.google.com/url?sa=t&rct=j&q=&esrc=s&source=web&cd=&ved=2ahUKEwiA5bjf54TqAhUIG80 KHeqPDusQFjACegQIAhAB&url=https%3A%2F%2Fwww.aph.gov.au%2FDocumentStore.ashx%3Fid%3D53 a5f52f-5ee9-414b-879b-6bbcdb7e8678%26subId%3D252459&usg=AOvVaw0CkVkNNIBpgeqhsEa HFCB

using data up to April 2020. Their studies thus focus on the *immediate* impact of the pandemic. Our study leverages the WBES-COVID 19 data that are mostly collected in June, July and August 2020. Therefore, we are able to capture the *medium-term* impact of the pandemic, and shed light on subsequent government policies such as the policy stringency of COVID-19 control policies. Moreover, the extent of firm survival and growth would clearly depend on the time length of the pandemic, and a medium-term impact study such as ours would be essential for understanding the overall impact on firms.

Empirically, we employ the WBES-COVID data that cover 18,770 firms in 36 countries to examine the effects of firm/country characteristics on firm survival and growth. We document four sets of findings. First, firm and country characteristics play an important role in private firms' survival and growth. Larger firms are much better equipped to weather the pandemic, as they exhibit lower risk of closure and higher revenue growth. However, larger firms also suffer higher employment loss. A possible explanation is that larger firms have more redundant and/or unproductive workers before the pandemic, and they take advantage of the crisis to shed them, which would be politically untenable during normal times. We also find that state ownership is linked to higher revenue and employment growth rates during the pandemic. State ownership has been linked to soft-budget constraints (Megginson, Ullah, and Wei, 2014). While the pandemic has brought financial shocks to all firms (Chundakkadan et al. 2020), firms with soft-budget constraints are more likely to survive and even thrive. Interestingly, subsidiaries affiliated with larger parent firms are also significantly more likely to survive and grow during the pandemic, possibly due to supports from their respective parent companies. Firms with foreign ownership are, in contrast, more likely to be closed. A likely explanation is that foreign firms are also more likely exporters, which suffer more from interruptions in global supply chains than nonexporters at the onset of the pandemic (Grassia et al. 2020; Peng and Kathuria, 2021). Firms with more financial obstacles are less likely to survive or grow their revenues. Firm survival and growth are, as expected, more likely in countries with a higher per capita income, a lower COVID spread, and a less stringent COVID control policy.

The second set of findings is that several firm/country characteristics affect firm growth very differently during normal versus the pandemic times. For example, state ownership and subsidiary status play a significant and positive role in firm growth in pandemic times, whereas they are not important determinants of firm growth in normal/pre-pandemic times—ownership and organizational features that cushion the shocks thus do help greatly during the pandemic.

Firm size also affects firm growth differently during normal versus pandemic times: larger firms are better able to maintain higher sales growth, while reducing their employment more aggressively.

The third set of findings is that the relationship between firm characteristics and firm survival/growth is significantly impacted by the stringency of a country's COVID control policy. For example, large firms are better equipped than small firms to withstand the increased stringency of government-imposed COVID measures. As governments raise their COVID policy stringency, firms with state ownership suffer higher revenue and employment losses, possibly due to their higher levels of compliance with government mandates.

The final set of findings is that as an informal institution, one national culture measure—long-term orientation—plays a positive role in firm survival and growth. As a formal institution, the overall country governance quality is positively linked to firm closure and negatively linked to firm sales and employment growth, perhaps reflecting the stronger enforcement of the lockdown policies.

Our paper endeavors to make two contributions. First, it adds to the literature on corporate immunity by being one of the first to comprehensively examine the effects of organizational resources, national institutions, and culture on the survival and growth of private firms during the pandemic across a large of number of mostly developing countries. Our paper is perhaps the first on the pandemic's medium-run effects on firms around the world. As mentioned earlier, our study differs from the existing literature in three aspects: our focus on unlisted/private firms, our cross-country setting, and our examination of the medium-run timeframe into the pandemic. Our paper enriches this fast increasing literature by starting to fill these gaps. Second, our paper is among the first to find that some firm/country characteristics impact firm survival and growth differently in normal versus in pandemic times. We report that a country's cultural characteristics, such as long-term orientation, are positively related with private firms' survival and growth, complementing similar findings among listed firms (Ding et al., 2021) and the previous literature on the importance of informal institutions on firm performance and strategic choices in developing countries (Duran et al. 2019). We also find that a country's governance quality is negatively associated with firm survival and growth during pandemic times, a finding that indicates governance plays a different role in facilitating firm prosperity during normal times and pandemic times.

The rest of the paper is organized as follows. Section 2 provides a review of the burgeoning corporate immunity literature, while Section 3 describes the data and variables used in the study. Section 4 reports our empirical results and Section 5 concludes our study.

2. Literature Review

The corporate immunity literature studies factors that help immunize firms from negative shocks such as the COVID-19 pandemic (Albuquerque et al. 2020; Ding et al. 2021). Here we review recent studies on firm-level and country-level determinants of corporate immunity.

2.1. Effects of Organizational Resources

Firm survival and growth depend on how firms leverage their bundle or organizational resources and capabilities (Penrose, 1959). During COVID-19, firms that possess such valuable resources are more likely to survive and grow. Specifically, firms whose business models can adjust well to social distancing or remote work are less negatively affected by the pandemic. Pagano et al. (2020) provide firm-level evidence from U.S. listed firms that those firms that adapt well to social distancing outperform their peers that require face-to-face contacts to function. Papanikolaou and Schmidt (2020) offer industry-level evidence that industries less equipped to work remotely suffer bigger declines in sales, employment, and stock returns. Bretscher et al. (2020) show that labor-intensive industries, which are less adaptive to remote work, exhibit more negative stock returns at the onset of the pandemic.

Part of the bundle of organizational resources, firm ownership and affiliations also play an important role in determining firms' effectiveness in coping with the pandemic. Bansal et al. (2020) document that affiliated firms outperform unaffiliated firms in India. Amore, Quarato and Pelucco (2021) report that family-controlled firms fare better than nonfamily-controlled firms in Italy. Ding et al. (2021) show that firms with hedge fund ownership are better immune to the pandemic than firms without such ownership.

Another aspect of the bundle of organizational resources, a firm's financial flexibility also appears to help weather the storm. Alfaro et al. (2020) and Ramelli and Wagner (2020) find that U.S. listed firms with lower leverage and higher cash holdings are linked to higher firm value. Fahlenbrach et al. (2020) show that financially flexible U.S. listed firms experience a smaller stock price decline at the onset of the pandemic than financially inflexible firms. Ding et al. (2021) examine listed firms around the world and document that high cash holdings, low

debt ratio, and high profitability help minimize firms' exposure from the pandemic-induced stock market shock.

Several studies suggest that corporate social responsibility (CSR) activities may help immunize firms in the crisis. On the one hand, Albuquerque et al. (2020) point out that U.S. listed firms with higher environmental, social, and governance (ESG) ratings outperform their peers with a lower ES rating in terms of stock returns and operating profit. Ding et al. (2021) examine listed firms from 56 countries and find that firms with more CSR activities suffer less from the pandemic-induced stock return decline. Bae et al. (2021) report that CSR positively affects stock returns during the crisis only if CSR activities are consistent with a firm's institutional environment. On the other hand, Demers et al. (2020) find that ESG scores have no significant relation with U.S. listed firms' stock returns at the onset of the pandemic. They argue that it is premature to conclude that ESG helps immunize firms in this crisis.

The literature suggests that a cohesive corporate culture—another crucial organizational resource—tends to help weather the pandemic storm better. Shan and Tang (2020) find that Chinese listed firms with more satisfied employees perform better during COVID-19 than those with less satisfied employees. Li et al. (2020) examine U.S. listed firms using the conference call data from January to April 2020 and find that firms with a strong corporate culture are more resilient to the pandemic shock.

All these studies examine listed firms in China, Italy, and the United States as well as around the world. A much smaller number of studies have examined private (unlisted) firms' behaviors and outcomes during the pandemic. Bloom et al. (2020, 2021) offer evidence from the United Kingdom and the United States that COVID-19 led to significant total factor productivity (TFP) reduction and sales losses. Dai et al. (2021a, 2021b) find Chinese SMEs (especially those that export) were severely adversely affected by the pandemic in terms of temporary and permanent closures. Gu et al. (2020) show that state-owned and/or foreign-owned firms in southern China suffer less than their nonstate-owned or nonforeign-owned peers in terms of daily output. Two studies find that firms tend to weather the pandemic storm better when facing less financial pressure: Carletti et al. (2020) study both listed and private firms in Italy and find that private firms and firms with a high pre-pandemic leverage are more likely to exhibit financial distress. Chundakkadan et al. (2020), using survey data of 13 countries from the World Bank, demonstrate that financially constrained firms are more likely to go bankrupt than non-financially constrained firms.

2.2. The Effects of National Institutions and Culture

The COVID-19 pandemic has inflicted pains upon every country in the world, the degree of which varies widely from country to country. Global variations are huge in terms of economic and institutional development (North, 1990), cultural characteristics (Hofstede, 2001), government COVID policy stringency (Fang et al. 2020), and COVID spread severity (Fang et al. 2020). This begs the question: Do country characteristics such as national institutions and culture affect firm behaviors and outcomes during the pandemic? Some scholars have started to examine this question.

As a logical starting point, at the onset of the pandemic, researchers have examined the relation between COVID infection rates and stock returns. They find a significantly negative link between them: Bretscher et al. (2020) on the relation between COVID infection rates in U.S. counties and stock returns of firms headquartered in respective counties. Similar findings are reported by Bansal et al. (2020) in India, and by Erdem (2020) in a cross-country setting.

As the pandemic progresses, countries around the world begin to adopt various COVID control policies with a wide degree of stringency. Ashraf (2020b) documents that strict social distancing policies negatively affect stock market returns, while government income support and testing policies are positively linked to stock market returns. Kaczmarek et al. (2020) find that countries with high stringency polices outperform countries with weak stringency polices in terms of stock market returns.

A few scholars have examined the role of a country's culture during the pandemic. Culture is defined as "the collective programming of the mind that distinguishes the members of one group or category of people from another" (Hofstede, 2001). Hofstede (2001) categorizes a country's culture into six dimensions—namely, power distance, individualism, masculinity, uncertainty avoidance, long-term orientation, and indulgence. Among the six, uncertainty avoidance, long-term orientation, and individualism are the most examined in the context of the pandemic. High uncertainty avoidance has been linked to larger stock market declines (Ashraf 2020a; Fernandez-Perez et al. 2021). Long-term orientation is commonly linked with smaller stock market declines, as long-term oriented investors can better understand the transient nature of the pandemic, and hence, are unlikely to overreact (Hofstede, 2020; Zaremba et al., 2021). Concerning the effect of individualism, some scholars find that more individualistic countries suffer smaller stock market declines (Fernandez-Perez et al., 2021), while others find the opposite (Kaczmarek et al., 2020). These studies mainly focus on the

effects of culture on stock markets. Our paper extends this line of research by examining the effects of culture on firm survival and growth during the pandemic.

Prior research has shown that democracy and institutional quality affect stock market returns, firm behavior, and performance (Beck et al., 2005; Hooper, Sim and Uppal, 2009; Xie and Li, 2018). The corporate immunity literature has not shown a consensus regarding the impact of democracy and institutional quality in the pandemic. Fernandez-Perez et al. (2021) and Fang et al. (2020) show that democratic governments and low political corruption are linked to smaller market declines and lower mortality rates during the pandemic. Erdem (2020) find that political and economic freedom, especially the freedom of expression, alleviate the negative relation between the severity of the pandemic and stock returns. However, Zaremba et al. (2021) find that autocratic governments with strong power are able to enforce stringent COVID policies to limit the negative impact of COVID-19. Ding et al. (2021) find that the power of media, proxied by the World Bank's measure of voice and accountability, has no significant impact on stock market reactions to the pandemic.

Overall, an emerging and rapidly accumulating stream of the literature hints at some possible relationship linking organizational resources, country institutions, and national culture to firm survival and growth during COVID. Reported next, our empirical efforts take a deep dive into this direction.

3. Data and Variables

3.1. The Sample

We obtain our sample of firms from two sources. The first firm-level data set is the World Bank Enterprise Surveys (WBES), which covers over 130 countries around the world. WBES covers firm characteristics, access to finance, employment, sales, costs, and various measures of business-government relations. WBES data have been widely used in economics, finance, international business, and management research.⁴ The second firm-level data set is the World Bank Enterprise Follow-up Surveys on COVID-19 (WBES-COVID), which represents follow-up surveys on WBES firms. WBES-COVID collects firm information on sales changes, business operations, labor adjustments, access to finance, expectations about the future, and support from the government in response to COVID-19. WBES-COVID now contains survey

⁴ Akins et al. (2017); Ayyagari et al. (2011, 2014); Barth et al. (2009); Beck et al. (2005, 2006, 2008); Cheng et al. (2020); Cull and Xu (2005); Jensen et al. (2010); Knack and Xu (2017); Houston et al. (2011); Liu et al. (2021); Xu (2011); and Zhou and Peng (2012).

data for 36 countries as to February 2021. We then merge WBES-COVID with WBES by the unique firm id to combine all firm-level information.

In addition to these two firm-level data sets, we utilize several country-level data sources. Country-level macro and economic variables are gathered from the World Development Indicators (WDI). National culture variables and country governance variables are obtained from Hofstede Insights and World Governance Indicators (WGI), respectively. The number of COVID-19 cases are gathered from Center for Systems Science and Engineering at Johns Hopkins University, while the government policy responses to COVID-19 are gathered from Oxford COVID-19 Government Response Tracker. The final sample used in this study includes 18,770 firm-observations from 36 countries.

3.2. Variables

3.2.1. Dependent Variables

The dependent variables in this study are firm closure and growth rates during the pandemic. Firm closure is measured by *Closure*, a dummy variable that equals one for firms that are temporarily or permanently closed, and zero otherwise. Firm growth is measured by the percentage changes in sales and in employment between the latest month (survey month) and the same month a year ago, denoted as *Sales growth* and *Employment growth*, respectively.

3.2.2. Firm-Level Characteristics

To evaluate how firm characteristics affect firm closure and growth during the pandemic, we include the following set of firm-level variables drawing on prior studies (Beck et al., 2005; Beck et al., 2008; Cheng et al., 2020; D'Souza et al., 2017; Liu et al., 2020; Ullah and Wei, 2017). We first include firm size, firm age, and three ownership variables. Firm size, Ln(firm size), is calculated as the natural logarithm of the number of employees. Firm age, Ln(firm age), is the natural logarithm of firm age plus one. Ownership is measured in three ways. (1) State ownership, State, is a dummy variable that equals one for firms with government/state ownership, and zero otherwise. (2) Foreign ownership, Foreign, is a dummy variable that equals one for firms with foreign ownership, and zero otherwise. (3) Subsidiary status, Subsidiary, is a dummy variable that equals one if a firm is a subsidiary of another firm, and zero otherwise.

Since publicly listed firms are generally larger and have better access to finance than private firms, we also examine if a firm is a publicly listed firm—*Public*, a dummy variable

that equals one, and zero otherwise. *Exporter* is included to indicate if a firm is an exporter. Lastly, we include a measure of financial constraint, *Finance obstacle*, a categorical variable on a scale 0-4, which measures the degree of obstacle in accessing financing, with 0, 1, 2, 3, and 4 indicating no, minor, moderate, major, and very severe obstacles, respectively.

3.2.3. Country Characteristics

We control for country-level economic development, proxied by *Ln(GDP per capita)*, the natural logarithm of GDP per capita in constant 2010 U.S. dollars. We further control for a demographic variable, *Age65*, the percentage of population of age 65 or older, as older population is more vulnerable to COVID-19 (Ding et al., 2021).

We examine the effects of three cultural dimensions on firm survival and growth during the pandemic. Specifically, we focus on three of Hofstede's six dimensions: (1) *Individualism* reflects the degree of interdependence that a society maintains among its members. (2) *Uncertainty avoidance* reflects the extent that members of a society feel uncomfortable with ambiguity and uncertainty. (3) *Long-term orientation* describes how a society encourages thrift/frugality and efforts in modern education to prepare for the future.

A country's governance quality may have a direct impact on the effectiveness of its responses to the COVID pandemic, hence affecting firm survival and growth. We employ the six commonly-used governance measures in Worldwide Governance Indicators (WGI) data set, constructed by Kaufman, Kraay and Mastruzzi (2011): namely, *Corruption control*, *Government effectiveness*, *Political stability*, *Regulatory quality*, *Rule of law*, and *Voice and accountability*. These measures take the values from -2.5 to 2.5, with a higher value indicating a higher governance quality. We also take the average of the six measures, *Average WGI index*, to proxy a country's overall governance quality (Karolyi, 2015; Pinkowitz et al., 2016).

Our data on COVID-19 cases are obtained from the Center for Systems Science and Engineering at Johns Hopkins University, which collects and maintains COVID-19 data from 192 countries/regions, including the number of confirmed cases, number of deaths, and number of recoveries, starting from January 22, 2020, and ongoing. Our COVID density measure, *COVID spread*, is constructed as the number of cumulative confirmed COVID-19 cases per 1,000 persons, one day before the WBES COVID-19 follow-up surveys begin in each country. We also employ Oxford COVID-19 Government Response Tracker (OxCGRT) database, which provides information on the government policy responses to the COVID-19 pandemic across more than 180 countries. The government COVID policy stringency index, *COVID*

policy stringency, is a comprehensive measure that takes into account nine response indicators, including school and workplace closures, cancellation of public events, restrictions on public gatherings, closures of public transport, stay-at-home orders, public publicity campaigns, restrictions on internal activities and international travel controls. The stringency index ranges from 0 to 100, with 100 indicating the strictest policy.

3.3. Summary Statistics

The descriptive statistics of firm survival and growth are reported in Table 1, Panel A. Among the sample firms, 11% are temporarily or permanently closed since the start of the pandemic. Compared with the same month (survey month) a year earlier, the firms on average experience 38.7% and 4.8% points decline in revenues and employees, respectively. Firm survival (i.e., 1 - *Closure*) and growth rates vary vastly among firms—see the very large corresponding standard deviations.

The summary statistics of the firm characteristics are in Table 1, Panel B. A median firm has 21 employees and is about 19 years old. About 1%, 12%, and 16% of firms have state ownership, foreign ownership, and a parent company, respectively. About 7% firms in our sample are publicly listed, and 33% are exporters. The median level of financial obstacle, *Finance obstacle*, is 1, indicating that about half of the firms in our sample perceive their financial obstacles as minor.

Table 2 provides the means of firm survival and growth rates by country. Firm survival rates (which are the reverse for the report closure rates) and growth rates during the pandemic vary tremendously across countries. Among the 36 countries, Chad reports the highest percent of firm closure (82%). Firms in El Salvador suffer the biggest sales decline (-59.9%), while those of Georgia report the deepest employment cut (-26.5%). On the more positive side, Estonia and Lithuania report the lowest percent of firm closure (i.e., without any), while Latvia even reports an increase in sales (1.1%) and employment (13.4%) during the pandemic.

The summary statistics of country-level variables are in Table 1, Panel C. The median per capita GDP in our sample countries is US\$11,968 in 2010 value. The average scores of *Individualism*, *Uncertainty avoidance*, and *Long-term orientation* are 41.5, 81.2, and 49.9, respectively. Both *COVID spread* and government *COVID policy stringency* vary markedly across our sample countries. Lastly, country governance quality measures also vary widely among the sample countries, as shown by the very large standard deviations.

Table 3 reports the Pearson correlation matrix for all our key variables. Consistent with our expectation, *COVID policy stringency* positively correlates with firm closure and negatively correlates with firm sales growth and employment growth in crisis. Most of the firm and country characteristics are also significantly correlated with firm closure and growth. We do not observe any correlation coefficient with an absolute value higher than 0.5, a rudimental check for possible multicollinearity issues in our regressions.

[Table 3 here]

4. Empirical Results

4.1. Effects of Firm and Country Characteristics

To assess the impact of firm and country characteristics on firm survival and growth during the pandemic, we estimate the following baseline cross-sectional regression:

$$Y_{ick} = \alpha + \beta \text{ Firm Characteristics}_{ick}$$
$$+ \gamma \text{ Country Characteristics}_c + v_k + \varepsilon_i \tag{1}$$

The subscripts i, c, k represent firm, country, and industry. Left-hand side of Eq. (1), Y, represents three dependent variables, namely, Closure, Sales growth, and Employment growth. Firm and country characteristics are the same as those described in Section 3.2 above and reported in Table 2 (Panel B and C). We also include the industry fixed effects, v_k , to absorb unobserved industry-level variables that may affect firm survival and growth during the COVID-19 pandemic. We report robust standard errors clustered at the country-industry level. To mitigate the effect of outliers, we winsorize all firm-level continuous variables at the 1st and 99th percentiles.

Before we present the results, we must acknowledge that what we find reflects correlations and descriptive patterns, not causality. Nevertheless, (1) the timeliness of the data, (2) the comprehensiveness of our country coverage, and (3) the urgency to understand the effects of this still ongoing pandemic on firms in countries and firms with different characteristics—all these factors combine to help make this study potentially useful to the research community and policy makers. To the extent we can, we have interpreted our results with insights from the literature so that our interpretations have some foundations.

Table 4 presents our baseline regression results. Several firm characteristics play as important role in firm survival and growth during the pandemic. In particular, larger firms are much less likely to face closure. Large firms are also more likely to grow their revenues during

the pandemic. One explanation is that smaller firms suffer a bigger negative shock to their revenues. As large firms, with stronger financial cushion and capacity, are better equipped to absorb the pandemic shock, they are able to take over the lost market share by small firms. However, larger firms exhibit higher employment loss. It could be that bigger firms have more redundant or unproductive workers before the pandemic. These firms then use the pandemic as a political cover to get rid of their excessive workers, which would be politically untenable during normal times. Relatedly, firms with less financial obstacles are more likely to survive or grow their revenues.

Ownership and parent-company affiliations matter a great deal during the pandemic. First, state ownership is associated with higher sales and employment growth rates during the pandemic. State ownership has been linked to soft-budget constraints (Megginson et al. 2014). While the pandemic has brought financial shocks to all firms (Chundakkadan et al. 2020), firms with soft-budget constraints are more likely to survive and even thrive, as they can always go to the government to ask for financial or other material supports. Besides profit consideration, state-owned firms are also more likely to have political considerations, especially when it comes to absorbing negative shocks to employment during a crisis. The small share of stateowned enterprises in our sample is suggestive that they tend to be those firms on the "commanding heights" that the country wants to hold tight control, and for such strategically important firms, the logic of efficiency often is not the highest priority (Huang et al. 2017). Second, firms with foreign ownership are more likely to face closure during the COVID-19 pandemic. Government bailouts often exclude foreign-owned firms. Foreign-owned firms are also more likely to be exporters, as shown by a highly significant correlation coefficient between foreign ownership and exporter status (Table 3). Exporting more, foreign-owned firms would then suffer more from pandemic interruptions with global supply chains and from export demand shocks (Grassia et al. 2020; Peng and Kathuria, 2021). Finally, thanks to their parent companies, subsidiaries are significantly more likely to survive and grow during the pandemic.

As expected, firms in countries with a higher per capita GDP, a lower COVID density, and a less stringent COVID control policy are more likely to survive and grow. *Ln(GDP per capita)* is negatively related with firm closure and positively related to sales and employment growth. Consistent with Ding et al. (2021), the results suggest that firms from relatively high income countries are much more likely to survive the pandemic and even thrive. Moreover, the association of *COVID policy stringency* with firm closure and revenue loss are both statistically and economically significant. One standard deviation increase in policy stringency is

associated with an increase in the possibility of firm closure by 4% (i.e., 0.002×20.1) and decreases firm revenue by 6.2% (i.e., -0.31×20.1).

[Table 4 here]

Prior research has shown that (non-listed) private firms tend to be smaller and face more obstacles in the business environment than publicly listed firms (Beck et al., 2005, 2006; Nikolov, Schmid, and Steri, 2020). Hence, private firms may behave differently from publicly listed firms during the pandemic. As a robustness test, we divide the full sample into publicly listed firms and private firms, and rerun the baseline regressions for each subsample. The results are reported in Table 5.

Firm size continues to have the most consistent and significant impact on firm survival and revenue growth across all firms, while firm age exhibits a negative impact on sales growth for public firms and negative impact on employment growth for private firms. Subsidiary status plays a highly positive and significant role in the survival of private firms during the pandemic, but is insignificant for the survival of public firms. Being a subsidiary firm is important for growth for both public and private firms, consistent with the baseline findings. Another notable observation is that financial constraint, *Finance obstacle*, is only significantly detrimental to the survival and revenue growth of private firms, whereas it plays no significant role in publicly listed firms' survival and growth during the pandemic. This is not surprising: listed firms have better access to financing via equity markets.

[Table 5 here]

4.2. The effects of firm/country characteristics during normal versus pandemic times

An advantage of our study over the extant literature on corporate immunity is that we employ the pre-pandemic WBES with the WBES-COVID surveys. We are able to combine both data sets via a firm's unique identifier. The combined data set contains firm-level information for both the pre-pandemic period ("normal times") and the pandemic period ("pandemic times"). This allows us to examine if, what, and how firm/country characteristics affect firm growth differently in normal versus pandemic times. Since by definition, the pandemic follow-up survey is conditional on firm existence in the last survey—that is, *Closure* is always zero for WBES—we do not analyze the closure outcome when comparing behavior in normal times and during the pandemic. Instead, we examine firm sales and employment growth in these two

periods. The regression results are reported in Table 6, with columns (1) and (3) for pandemic times, and (2) and (4) for normal times.

Most of the firm characteristics affect firm growth differently in pandemic versus in normal times. In particular, large firms exhibit employment reduction during the pandemic (column (3)) but an increase in normal times (column (4). These results provide additional support for the notion that large firms may have taken advantage of the crisis to shed themselves of excessive workers, which may be politically untenable during normal times. Large firms are able to have significantly higher growth in sales during the pandemic times versus the normal times. While firm age has no effect on sales growth during the pandemic (column (1)), it is negatively related to sales growth in normal times (column (2)). Firm age is slightly positively related to employment growth during the pandemic (column (3)), but significantly negatively related to employment growth in normal times (column (4)). These results may suggest that older firms are better equipped to weather the pandemic shock than younger firms in terms of protecting their workers. An explanation here is that older firms have an employee force that have been together for longer, and have stronger firm culture and thus the owners/managers feel stronger normative pressure, as suggested by the sociology literature (Scott 2014: 64), to maintain employment.

Favorable ownership and parent-company affiliations seem to better cushion firms during pandemic times as opposed to normal times. Ownership structure is also associated with firm growth differently in pandemic versus in normal times. Notably, state ownership plays positive and highly significant role in both sales (column (1)) and employment (column (2)) growth during the pandemic, providing further support for the notion that government ownership and the associated political connections are useful during pandemic times. However, in normal times, state ownership has neither impact on sales growth (column (2)) nor on employment growth (column (4)). Furthermore, being a subsidiary is also highly useful to firm growth during the pandemic (columns (1) and (3)), but not so in normal times (columns (2) and (4)).

Not unexpectedly, country-level economic development, proxied by *Ln(GDP per capita)*, has a positive and significant association with sales growth in pandemic times (column (1)), but an insignificant association with sales growth in normal times. A possible explanation is that during the pandemic, domestic demand, which rises with GDP per capita, becomes more important for firm growth (Peng and Kathuria, 2021).

4.3. The interactive effect of COVID policy stringency

Our baseline results in Table 4 show that both firm characteristics and COVID policy stringency have significant effects on firm survival and growth during the pandemic. We now examine the interactive effects of firm characteristics and COVID policy stringency on firm survival and growth. Specifically, we estimate the following regression model:

$$Y_{ick} = \alpha + \beta Firm \ characteristics_{ick} + \delta Firm \ characteristics_{ick} \times \\ Covid \ policy \ stringency_c + \gamma Country \ characteristics_c + v_k + \varepsilon_i \qquad (2)$$

The dependent variables include the same three as in our baseline model. Our focus is the coefficients of the interaction term, $Firm\ characteristics_{ick} \times Covid\ policy\ stringency_c$, as they will provide insights into how a government's COVID policy affects the relation between firm characteristics and firm survival and growth. The results are reported in Table 7.

Government COVID policy stringency has significant interactions with some of the firm characteristics in shaping firm survival/growth relations. When governments increase their COVID control policy stringency: (1) large firms are better positioned than small firms to survive and even grow their revenues; (2) firms with state ownership exhibit higher revenue loss, possibly because state-owned firms, typically being subject to stronger government influence, are more amendable to government mandates.

When governments increase their COVID control policy stringency, subsidiary firms appear to be better equipped than non-subsidiaries to protect their workers' employment or even add to their labor force. This further suggests possible material help from parent companies in times of crisis. In terms of COVID policy stringency's impact on sales, publicly listed firms are also better positioned than private firms to withstand an increased COVID policy stringency, possibly due to public firms having more resources, financial or otherwise, to cope with the sustained lockdowns and social distancing.

[Table 7 here]

4.4. The effects of country culture and governance

COVID-19 again highlights the importance of country-level formal and informal institutions in handling the crisis. During the pandemic, some of the formal institutions would stop to work due to the stay-at-home restrictions, among others. Then informal institutions would loom large

to shape organizational choices (North, 1990; Peng, 2003). We thus now turn our attention to how a country's culture (i.e., informal institutions) and governance (i.e., formal institutions) affect firms' survival and growth during the pandemic.

4.4.1. The effects of national culture

We focus on three of Hofstede's (2001) six culture dimensions: individualism, uncertainty avoidance, and long-term orientation. These three dimensions have been found to be more impactful than other culture dimensions on firm behaviors and outcomes during the COVID-19 pandemic (Ashraf, 2020; Fernandez-Perez et al. 2020; Gorodnichenko and Roland, 2020; Kaczmarek et al. 2020; Zaremba et al. 2021). We add the three culture dimension measures to our baseline model (Eq. (1)), rerun the regressions, and report the results in Table 8.

On the one hand, *Long-term orientation* is significantly negatively related with firm *Closure* and positively related with revenue and employment growth. This result indicates that firms are more likely to survive and grow in more long-term orientated countries, which is consistent with Zaremba et al. (2021). Duran et al. (2019) also find that long-term orientation of a country, which they classify as an informal enabling institution, affects the relative performance of publicly listed family versus non-family firms. It is likely that during the dark times with no light at the end of the tunnel known as COVID-19, firms (and managers) in countries with a higher level of *Long-term orientation* may have more enduring patience, stronger resilience, and more powerful optimism, whereas firms (and managers) in countries with a low level of *Long-term orientation* may be more easily discouraged to close down and declare bankruptcy.

On the other hand, neither *Uncertainty avoidance* nor *Individualism* has a significant association with firm survival or growth during the pandemic. Their impact thus awaits further research.

[Table 8 here]

4.4.2. The effects of governance quality

We employ the WGI to assess the effects of country governance quality on firm survival and growth during the pandemic. We examine all six dimensions of a country's governance quality, namely, corruption control, government effectiveness, political stability, regulatory quality, rule of law, and voice and accountability. Following standard practice, we also use the average value of the above six measures, *Average WGI index*, to proxy a country's overall governance

quality (Kaufman, Kraay and Mastruzzi 2011). Table 9 reports the results of the effects of country governance on firm survival and growth during the pandemic.

Robustly, higher country governance quality is associated with higher odds of firm closure and lower sales and employment growth during the pandemic. In particular, firms in less corrupt countries (better corruption control) face higher probability of closure, bigger loss of revenues and employment (Panel A). Firms in countries with more effective governments suffer bigger employment loss (Panel B). Firms in countries where enforcement of regulation is more effective suffer higher odds of closure and bigger losses in revenue and employment (Panel D), while firms in better rule of law countries face higher odds of closure and suffer bigger employment loss (Panel E). Interestingly, as shown in Panel F, firms in countries with better citizen political participation and more accountable government are also more likely to face closure and suffer bigger revenue and employment losses. Using the average value of the above six measures, overall we find that high country governance quality is associated with worse firm survival and growth during the pandemic.

These findings may be counter-intuitive in normal times. However, during a pandemic, these findings are plausible and logical. In countries with highly effective governments and less corruption, citizens are more likely to entrust the governments to carry out COVID-19 control policies. Even if some citizens are not happy with the lockdowns or other mitigation measures, highly effective governments are better able to mobilize institutional resources to enforce rules and regulations during the pandemic. The upshot? More complete and more enduring lockdowns. In the short run, firms in these countries suffer as a consequence.

[Table 9 here]

5. Conclusion

This study probes the effects of organizational resources, country institutions, and national culture on firm survival and growth during the still ongoing COVID-19 pandemic. Analyzing the combined WBES and WBES-COVID data sets, we document a number of findings. At the firm level, large firms are much more likely to survive and even thrive in terms of growing their revenues during the pandemic. However, large firms experience bigger employment loss than small firms. Our findings about firm size in the pandemic times compliment Beck et al. (2005), who find that firm size also plays a significant role in firm growth during normal times. Reflecting favorable access to resources, state ownership and subsidiary status consistently play a useful role in weathering the negative pandemic shock. Firms in high income countries

are also better equipped to survive and even thrive during the pandemic. We further find that most firm characteristics impact publicly listed firms and private firms quite differently during the pandemic. For example, subsidiary status is useful in reducing the odds of closure only among private firms. Financial constraint is detrimental to firm survival and revenue growth only among private firms.

Leveraging the advantage of combining the WBES and WBES-COVID data sets, we are able to compare and contrast the effects of organizational resources on firm growth in pandemic versus in normal times. Indeed, most organizational resources affect firm growth differently in pandemic versus in normal times. Large firms exhibit employment reduction during the pandemic but an increase in normal times, supporting the notion that large firms might have taken advantage of the crisis to shed themselves of excessive workers, which may be politically untenable during normal times. Large firms have higher growth in sales during the pandemic times versus the normal times. Older firms are better able to weather the pandemic shock than younger firms in protecting their workers, perhaps because older firms have an employee force that has been together for longer, and have stronger firm culture so that the owners/managers feel stronger normative pressure to maintain employment. Favorable ownership and parent-company affiliations seem to better protect firm growth during the pandemic times versus normal times. Richer countries are better able to preserve firm growth during the pandemic but not in normal times.

At the country level, we find that government COVID policy stringency plays an important role in the relations between firm characteristics and firm survival and growth, and larger firms and firms with parent companies tend to be less harmed by stringent COVID policy. A country's cultural characteristics are also important determinants of firm survival and growth in the pandemic. In particular, firms in countries with a long-term orientation culture are much better equipped to weather the negative shock of the pandemic. Interestingly, we observe consistently that high country governance quality leads to low odds of firm survival and slow revenue and employment growth during the pandemic.

In conclusion, organizational immunity during COVID-19 is likely driven by organizational resources, country institutions, and national culture—with various combinations. Their impact on firm survival and growth differs between normal (pre-pandemic) times and pandemic times. Given the worldwide efforts to tame COVID-19, it is clear that our learning about them is not going to stop any time soon.

References

- Akins, B., Dou, Y., & Ng, J. (2017). Corruption in bank lending: The role of timely loan loss recognition. *Journal of Accounting and Economics*, 63(2-3), 454-478.
- Al-Awadhi, A. M., Alsaifi, K., Al-Awadhi, A., & Alhammadi, S. (2020). Death and contagious infectious diseases: Impact of the COVID-19 virus on stock market returns. *Journal of Behavioral and Experimental Finance*, 27, 100326.
- Albuquerque, R., Koskinen, Y., Yang, S., & Zhang, C. (2020). Resiliency of environmental and social stocks: An analysis of the exogenous COVID-19 market crash. *The Review of Corporate Finance Studies*, 9(3), 593-621.
- Alfaro, L., Chari, A., Greenland, A. N., & Schott, P. K. (2020). Aggregate and firm-level stock returns during pandemics, in real time (No. w26950). *NBER Working Paper 26950*.
- Amore, M. D., Quarato, F., & Pelucco, V. (2021). Family ownership during the COVID-19 pandemic. *Available at SSRN 3773430*.
- Ashraf, B. N. (2020a). Stock markets' reaction to COVID-19: Moderating role of national culture. *Finance Research Letters*, 101857.
- Ashraf, B. N. (2020b). Economic impact of government interventions during the COVID-19 pandemic: International evidence from financial markets. *Journal of Behavioral and Experimental Finance*, 27, 100371.
- Ayyagari, M., Demirgüç-Kunt, A., & Maksimovic, V. (2011). Firm innovation in emerging markets: The role of finance, governance, and competition. *Journal of Financial and Quantitative Analysis*, 46(6), 1545-1580.
- Ayyagari, M., Demirgüç-Kunt, A., & Maksimovic, V. (2014). Bribe payments and innovation in developing countries: Are innovating firms disproportionately affected? *Journal of Financial and Quantitative Analysis*, 49(1), 51-75.
- Bae, K. H., El Ghoul, S., Gong, Z. J., & Guedhami, O. (2021). Does CSR matter in times of crisis? Evidence from the COVID-19 pandemic. *Journal of Corporate Finance*, 67, 101876.
- Bansal, A., Gopalakrishnan, B., Jacob, J., & Srivastava, P. (2020). When the Market Went Viral: COVID-19, Stock Returns, and Firm Characteristics. Stock Returns, and Firm Characteristics. *Working paper*.
- Barth, J. R., Lin, C., Lin, P., & Song, F. M. (2009). Corruption in bank lending to firms: Cross-country micro evidence on the beneficial role of competition and information sharing. *Journal of Financial Economics*, 91(3), 361-388.
- Beck, T., & Demirguc-Kunt, A. (2006). Small and medium-size enterprises: Access to finance as a growth constraint. *Journal of Banking & Finance*, 30(11), 2931-2943.
- Beck, T., Demirgüç-Kunt, A. S. L. I., & Maksimovic, V. (2005). Financial and legal constraints to growth: does firm size matter? *The Journal of Finance*, 60(1), 137-177.
- Beck, T., Demirgüç-Kunt, A., & Maksimovic, V. (2008). Financing patterns around the world: Are small firms different? *Journal of Financial Economics*, 89(3), 467-487.
- Bloom, N., Bunn, P., Mizen, P., Smietanka, P., Thwaites, G. (2020). The impact of COVID-19 on productivity. *NBER Working Paper 28233*.
- Bloom, N., Fletcher, R.S., Yeh, E. (2021). The impact of COVID-19 on U.S. Firms. *NBER Working Paper 28314*.

- Bretscher, L., Hsu, A., Simasek, P., & Tamoni, A. (2020). COVID-19 and the cross-section of equity returns: Impact and transmission. *The Review of Asset Pricing Studies*, 10(4), 705-741.
- Carletti, E., Oliviero, T., Pagano, M., Pelizzon, L., & Subrahmanyam, M.G. (2020). The COVID-19 shock and equity shortfall: Firm-level evidence from Italy. *The Review of Corporate Finance Studies*, 9(3), 534-568.
- Cheng, C. A., Sun, W., Ye, K., & Zhang, N. (2020). The effect of auditing on promoting exports: evidence from private firms in emerging markets. *Management Science*, 66(4), 1692-1716.
- Chundakkadan, R., Raj, R., & Sasidharan, S. (2020). Small Firms amidst COVID-19: Financial Constraints and Role of Government Support. *Available at SSRN 3691564*.
- Cull, R., Xu, L.C. (2005). Institutions, Ownership, and Finance: The Determinants of Investment among Chinese Firms. *Journal of Financial Economics*, 77, 117-146.
- Dai, R., Feng, H., Hu, J., Jin, Q., Li, H., Wang, R., Wang, R., Xu, L., Zhang, X. (2021a). The impact of COVID-19 on small and medium-sized enterprises (SMEs): Evidence from two-wave phone surveys in China. *China Economic Review*, 67, 1-16.
- Dai, R., Mookherjee, D., Quan, Y., & Zhang, X. (2021b). Industrial clusters, networks and resilience to the Covid-19 shock in China. *Journal of Economic Behavior & Organization*, 183, 433-455.
- Demers, E., Hendrikse, J., Joos, P., & Lev, B. (2020). ESG didn't immunize stocks against the COVID-19 market crash. *Available at SSRN 3675920*.
- Ding, W., Levine, R., Lin, C., & Xie, W. (2021). Corporate immunity to the COVID-19 pandemic. *Journal of Financial Economics*.
- D'Souza, J., Megginson, W. L., Ullah, B., & Wei, Z. (2017). Growth and growth obstacles in transition economies: Privatized versus de novo private firms. *Journal of Corporate Finance*, 42, 422-438.
- Duran, P., van Essen, M., Heugens, P.P.M.A.R., Kostova, T., & Peng, M.W. (2019). The impact of institutions on the competitive advantage of publicly listed family firms in emerging markets. *Global Strategy Journal* 9(2), 243-274.
- Erdem, O. (2020). Freedom and stock market performance during COVID-19 outbreak. *Finance Research Letters*. Forthcoming.
- Fahlenbrach, R., Rageth, K., & Stulz, R. M. (2020). How valuable is financial flexibility when revenue stops? Evidence from the COVID-19 crisis (No. w27106). *National Bureau of Economic Research*.
- Fang, S., Peng, M.W., Xu, L.C., & Yi, Y. (2020). The Interplay of Policy, Institutions, and Culture in the Time of COVID-19. *World Bank Policy Research Working Paper 9470*.
- Fernandez-Perez, A., Gilbert, A., Indriawan, I., & Nguyen, N. H. (2021). COVID-19 pandemic and stock market response: A culture effect. *Journal of Behavioral and Experimental Finance*, 29, 100454.
- Gorodnichenko, Y., & Roland, G. (2020). Culture, institutions, and democratization. *Public Choice*, 1-31.
- Grassia, M., Mangioni, G., Schiavo, S., & Traverso, S. (2020). (Unintended) Consequences of export restrictions on medical goods during the COVID-19 pandemic. *arXiv* preprint *arXiv*:2007.11941.

- Gu, X., Ying, S., Zhang, W., & Tao, Y. (2020). How do firms respond to COVID-19? First evidence from Suzhou, China. *Emerging Markets Finance and Trade*, *56*(10), 2181-2197.
- Halling, M., Yu, J., & Zechner, J. (2020). How Did COVID-19 Affect Firms' Access to Public Capital Markets?. *The Review of Corporate Finance Studies*, 9(3), 501-533.
- Hofstede, G. (2001). Culture's consequences: Comparing values, behaviors, institutions and organizations across nations. Sage publications.
- Hofstede, G. J. (2020). BOSS Blog 9: Corona across Cultures.
- Hooper, V., Sim, A. B., & Uppal, A. (2009). Governance and stock market performance. *Economic Systems*, 33(2), 93-116.
- Houston, J. F., Lin, C., & Ma, Y. (2011). Media ownership, concentration and corruption in bank lending. *Journal of Financial Economics*, 100(2), 326-350.
- Huang, Z., Li, L., Ma, G., & Xu, L.C. (2017). Hayek, local information, and commanding heights: Decentralizing state-owned enterprises. *American Economic Review*, 107(8), 2455-2478.
- Jensen, N. M., Li, Q., & Rahman, A. (2010). Understanding corruption and firm responses in cross-national firm-level surveys. *Journal of International Business Studies*, 41(9), 1481-1504.
- Kaczmarek, T., Perez, K., Demir, E., & Zaremba, A. (2020). How to survive a pandemic: The corporate resiliency of travel and leisure companies to the COVID-19 outbreak. *Tourism Management*, 84, 104281.
- Karolyi, G. A. 2015. Cracking the emerging markets enigma. New York: Oxford University Press.
- Kaufmann, D., A. Kraay, M. Mastruzzi. (2011). "The worldwide Governance Indicators: Methodology and Analytical Issues." *Hague Journal on the Rule of Law* 3(2), 220-246.
- Knack, S., Xu, L.C. (2017). "Unbundling Institutions for External Finance: Worldwide Firm-Level Evidence." *Journal of Corporate Finance* 44(C), 215-232.
- Li, K., Liu, X., Mai, F., & Zhang, T. (2020). The role of corporate culture in bad times: Evidence from the COVID-19 pandemic. *Available at/SSRN 3632395*.
- Liu, T., Liu, Y., Ullah, B., Wei, Z., & Xu, L.C. (2021). The dark side of transparency in developing countries: The link between financial reporting practices and corruption. *Journal of Corporate Finance*, 66, 101829.
- Liu, Y., Sah, N., Ullah, B., & Wei, Z. (2020). Financing patterns in transition economies: Privatized former SOEs versus ab initio private firms. *Emerging Markets Review*, 43, 100680.
- Liu, Y., Wei, S., & Xu, J. (2021). COVID-19 and Women-Led Businesses around the World. *Finance Research Letters*, 102012.
- Megginson, W. L., Ullah, B., & Wei, Z. (2014). State ownership, soft-budget constraints, and cash holdings: Evidence from China's privatized firms. *Journal of Banking & Finance*, 48, 276-291.
- Nikolov, B., Schmid, L., & Steri, R. (2021). The sources of financing constraints. *Journal of Financial Economics*, 139(2), 478-501.
- North, D.C. (1990). *Institutions, Institutional Change and Economic Performance*. Cambridge, UK: Cambridge University Press.
- Pagano, M., Wagner, C., & Zechner, J. (2020). Disaster resilience and asset prices. arXiv preprint arXiv:2005.08929.
- Papanikolaou, D., & Schmidt, L. D. (2020). Working remotely and the supply-side impact of COVID-19 (No. w27330). *NBER Working Paper 27330*.

- Peng, M.W. (2003). Institutional Transitions and Strategic Choices. *Academy of Management Review*, 28(2), 275-296.
- Peng, M.W., & Kathuria, N. (2021). COVID-19 and the scope of the firm. *Journal of Management Studies*, doi:10.1111/joms.12699.
- Penrose, E.T. (1959). A theory of the growth of the firm. New York: Wiley.
- Pinkowitz, L., Stulz, R. M., & Williamson, R. (2016). Do US firms hold more cash than foreign firms do? *The Review of Financial Studies*, 29(2), 309-348.
- Ramelli, S., & Wagner, A. F. (2020). Feverish stock price reactions to COVID-19. *The Review of Corporate Finance Studies*, 9(3), 622-655.
- Scott, W.R. 2014. *Institutions and Organizations: Ideas, Interests, and Identities*. Thousand Oaks, CA: Sage Publications.
- Shan, C., & Tang, D. Y. (2020). The value of employee satisfaction in disastrous times: Evidence from COVID-19. *Available at SSRN 3560919*.
- Ullah, B., & Wei, Z. (2017). Bank financing and firm growth: Evidence from transition economies. *Journal of Financial Research*, 40(4), 507-534.
- Xie, Z., & Li, J. (2018). Exporting and innovating among emerging market firms: The moderating role of institutional development. *Journal of International Business Studies*, 49(2), 222-245.
- Xu, L.C. 2011. "The Effects of Business Environments on Development: A Survey of New Firm-Level Evidence," *World Bank Research Observer* 26 (2), 310-340.
- Zaremba, A., Kizys, R., Tzouvanas, P., Aharon, D. Y., & Demir, E. (2021). The quest for multidimensional financial immunity to the COVID-19 pandemic: Evidence from international stock markets. *Journal of International Financial Markets, Institutions and Money*, 71, 101284.
- Zhou, J. Q., & Peng, M. W. 2012. Does bribery help or hurt firm growth around the world? *Asia Pacific Journal of Management*, 29(4): 907-921.

Table 1. Summary Statistics.

This table presents summary statistics for our dependent variables (Panel A), firm-level characteristics (Panel B) and country-level characteristics (Panel C). Please refer to the Appendix for variable definitions and data sources.

	N	Mean	Std	Min	Median	Max
	A. Closure	and Growt	h			
Closure t	14376	0.11	0.32	0	0	1
Sales growth t	10028	-38.70	32.63	-100	-40	50
Employment growth t	13511	-4.75	29.96	-130.70	0	91.57
	B. Firm-le	vel Charact	eristics			
Ln (firm size) _{t-1}	18770	3.30	1.33	1.61	3.05	6.81
Ln (firm age) _{t-1}	18560	2.88	0.70	0	2.94	5.31
State t-1	18489	0.01	0.12	0	0	1
Foreign t-1	18465	0.12	0.32	0	0	1
Subsidiary t-1	18756	0.16	0.36	0	0	1
Public t-1	18566	0.07	0.26	0	0	1
Exporter t-1	18469	0.33	0.47	0	0	1
Finance obstacle t-1	18386	1.24	1.24	0	1	4
	C. Country	Character	istics			
COVID policy stringency t	18179	54.45	20.10	16.67	54.63	96.30
COVID spread t	18770	2.28	2.87	0.02	1.13	14.14
Ln (GDP per capita) _{t-1}	18770	9.04	1.02	6.33	9.39	10.48
Age65 t-1	18770	14.34	6.82	2.12	15.20	23.01
Individualism t-1	16699	41.50	18.05	6	39	80
Uncertainty avoidance t-1	16699	81.21	15.61	50	85	100
Long-term orientation t-1	16050	49.99	22.77	14	52	82
Control of corruption t-1	18770	-0.09	0.64	-1.42	-0.06	1.54
Government effectiveness t-1	18770	0.17	0.66	-1.57	0.15	1.18
Political stability t-1	18770	0.11	0.68	-1.64	0.29	1.13
Regulatory quality t-1	18770	0.30	0.72	-1.46	0.51	1.59
Rule of law t-1	18770	0.02	0.72	-1.28	0.14	1.28
Voice and accountability t-1	18770	0.12	0.80	-1.42	0.22	1.24
Average WGI index t-1	18770	0.11	0.66	-1.36	0.24	1.24

Table 2. Firm Closure, Growth, and Macro Variables' Distribution by Country in Pandemic Times.

This table reports by country our dependent variables, i.e., Closure, Sales growth, and Employment growth. The table also reports country characteristics, including cultural characteristics, government COVID policy stringency, and COVID spread severity, and overall country governance quality. Please refer to the Appendix for variable definitions and data sources.

Country	Obs.	Closure t	Sales growth t	Employment growth t	Individualism t-1	Uncertainty avoidance t-1	Long-term orientation t-1	COVID policy stringency t	COVID spread t	Ave. WGI index t-1
Albania	369	13.53%	-51.20%	-2.02%	20	71	61	71.30	0.42	-0.08
Belarus	589	4.25%	-23.54%	-2.66%	25	95	81	16.67	7.28	-0.45
Bulgaria	765	8.48%	-31.82%	-9.79%	30	85	69	36.11	1.10	0.28
Chad	147	82.35%	-47.36%	-9.29%				74.07	0.05	-1.36
Croatia	397	2.91%	-26.06%	-4.27%	33	80	58	35.19	2.81	0.46
Cyprus	224	6.29%	-42.95%	-2.08%				76.85	0.79	0.83
Czech Republic	500	1.74%	-20.22%	-2.43%	58	74	70	38.89	2.80	0.93
El Salvador	711	25.81%	-59.97%	-0.97%	19	94	20	96.30	0.49	-0.29
Estonia	354	0.37%	-12.54%	0.51%	60	60	82	25.93	2.76	1.24
Georgia	600	23.64%	-56.23%	-26.52%	41	85	38	69.44	0.21	0.45
Greece	585	4.05%	-33.13%	1.36%	35	100	45	54.63	0.27	0.39
Guatemala	331	17.62%	-51.49%	-3.92%	6	98		96.30	0.88	-0.62
Guinea	149	28.85%	-56.75%	-7.62%				73.15	0.36	-0.87
Honduras	318	32.10%	-57.04%	-2.07%	20	50		96.30	1.43	-0.67
Hungary	795	1.93%	-21.05%	-0.81%	80	82	58	50.00	0.86	0.42
Italy	752	18.93%	-45.99%	-3.35%	76	75	61	63.89	3.82	0.56
Jordan	569	16.73%	-56.66%	-11.70%	30	65	16	48.15	0.12	-0.10
Latvia	334	6.41%	1.07%	13.43%	70	63	69	32.41	1.24	0.86
Lebanon	506	12.50%		-18.83%	40	50	14	57.41	14.14	-0.90

Table 2. Firm Closure, Growth, and Macro Variables' Distribution by Country in Pandemic Times. (Cont'd)

Country	Obs.	Closure t	Sales growth t	Employment growth t	Individualism t-1	Uncertainty avoidance t-1	Long-term orientation t-1	COVID policy stringency t	COVID spread t	Ave. WGl index _{t-1}
Lithuania	352	0.47%	-19.78%	-6.55%	60	65	82	37.04	1.86	0.96
Malta	241	2.56%	-32.43%	3.65%	59	96	47		4.68	0.87
Moldova	337	13.48%	-57.26%	-14.31%	27	95	71	80.56	2.31	-0.31
Mongolia	347	18.42%	-42.43%	-5.34%				60.19	0.09	0.01
Morocco	1082	18.33%	-50.35%	-8.82%	46	68	14	64.81	0.45	-0.29
Nicaragua	326	8.47%	-45.46%	-6.01%				18.52	0.28	-0.98
Niger	148	17.14%	-56.27%	-15.54%				25.93	0.04	-0.75
North Macedonia	350	3.17%	-34.21%	-9.28%	22	87	62		9.02	-0.03
Poland	1289	8.97%	-27.17%	-0.74%	60	93	38	39.81	1.13	0.65
Portugal	1062	10.24%	-30.22%	0.70%	27	99	28	56.94	5.87	1.07
Romania Russian	795	2.51%	-24.24%	-3.63%	30	90	52	42.59	3.37	0.24
Federation	1318	8.01%	-31.99%	-1.97%	39	95	81	74.54	2.93	-0.58
Slovak Republic	427	4.17%	-25.72%	-2.40%	52	51	77	28.70	0.87	0.71
Slovenia	399	0.81%	-22.76%	-1.15%	27	88	49	39.81	0.53	0.99
Togo	143	25.49%	-50.36%	-11.55%				53.70	0.07	-0.74
Zambia	572	21.50%	-50.60%	-12.08%	35	50	30	39.81	0.08	-0.45
Zimbabwe	587	12.27%	-52.10%	0.42%				70.37	0.02	-1.20

Table 3. Pearson Correlation Matrix.

	1	2	3	4	5	6	7	8	9	10	11	12	13	14
1. Closure t														
2. Sales growth t	-0.33ª													
3. Employment growth t	-0.14^{a}	0.21^{a}												
4. Ln (firm size) _{t-1}	-0.13^{a}	0.16^{a}	-0.13^{a}											
5. Ln (firm age) _{t-1}	-0.10^{a}	0.08^{a}	0.02^{b}	0.24^{a}										
6. State _{t-1}	-0.02^{b}	0.05^{a}	0.00	0.12^{a}	0.07^{a}									
7. Foreign _{t-1}	0.00	0.02^{c}	-0.03ª	0.28^{a}	0.00	0.06^{a}								
8. Subsidiary t-1	-0.03^{a}	0.04^{a}	0.04^{a}	0.18^{a}	0.08^{a}	0.04^{a}	0.17^{a}							
9. Public _{t-1}	-0.03ª	0.04^{a}	-0.01	0.23^{a}	0.13^{a}	0.27^{a}	0.11^{a}	0.10^{a}						
10. Exporter t-1	-0.06^{a}	0.10^{a}	-0.01	0.35^{a}	0.13^{a}	0.04^{a}	0.22^{a}	0.04^{a}	0.09^{a}					
11. Finance obstacle t-1	0.09^{a}	-0.13a	-0.02^{a}	-0.07^{a}	-0.04^{a}	-0.01	-0.05a	0.00	0.00	-0.06^{a}				
12. COVID policy stringency t	0.15^{a}	-0.26^{a}	-0.02^{a}	-0.02^{a}	-0.01 ^b	-0.08^{a}	-0.03a	0.08^{a}	-0.02^{a}	-0.12^{a}	0.14^{a}			
13. COVID spread t	-0.08^{a}	0.19^{a}	-0.02^{c}	0.03^{a}	0.12^{a}	0.07^{a}	-0.08^{a}	-0.09^{a}	0.07^{a}	0.04^{a}	-0.01	-0.12a		
14. Ln (GDP per capita) _{t-1}	-0.19^{a}	0.31^a	0.10^{a}	0.05^{a}	0.12^{a}	-0.04^{a}	-0.10^{a}	-0.09^{a}	-0.06^{a}	0.15^{a}	-0.20^{a}	-0.27^{a}	0.24^{a}	
15. Age65 t-1	-0.18a	0.31a	0.09^{a}	0.06^{a}	0.07^{a}	-0.02a	-0.08^{a}	-0.07a	-0.02ª	0.16^{a}	-0.23a	-0.34a	0.21a	0.91ª

Superscripts a, b, c represent significance level of 1%, 5%, and 10%, respectively.

Table 4. The effects of firm/country characteristics on firm survival/growth in pandemic times.

This table presents the effects of firm and country characteristics on firm survival or growth in pandemic times. Columns (1), (2), and (3) report results for our three dependent variables, i.e., Closure t, Sales growth t, and Employment growth t, respectively. The explanatory variables are a set of firm characteristics and a set of country characteristics. The sources and detailed variable definitions are shown in the Appendix. Robust t-statistics based on standard errors clustered at the country-industry level are reported in brackets. *, **, and *** indicate significance at the 10%, 5%, and 1% levels, respectively.

	Closure t	Sales growth t	Employment growth t
	(1)	(2)	(3)
Ln (firm size) _{t-1}	-0.027***	3.423***	-4.014***
	(-7.04)	(8.20)	(-7.25)
Ln (firm age) _{t-1}	-0.019**	-0.799	1.204*
	(-2.53)	(-1.11)	(1.68)
State t-1	-0.002	7.816***	5.918***
	(-0.11)	(3.76)	(2.96)
Foreign t-1	0.018**	-0.262	1.158
	(2.03)	(-0.18)	(1.21)
Subsidiary t-1	-0.033***	3.706***	5.482***
	(-3.31)	(2.94)	(5.46)
Public t-1	0.004	-0.012	2.054
	(0.36)	(-0.01)	(1.63)
Exporter t-1	0.012	-1.441	0.743
	(1.39)	(-1.53)	(0.79)
Finance obstacle t-1	0.010***	-1.194***	-0.255
	(3.33)	(-4.13)	(-0.92)
COVID policy stringency t	0.002***	-0.310***	-0.003
	(5.31)	(-7.56)	(-0.08)
COVID spread t	0.000	0.167	-0.525*
	(0.07)	(0.44)	(-1.71)
Ln (GDP per capita) _{t-1}	-0.054***	6.753***	4.298**
	(-2.78)	(4.04)	(2.37)
Age65 t-1	0.002	0.123	-0.165
	(0.73)	(0.46)	(-0.56)
Industry FE	YES	YES	YES
Observations	13081	9062	12323
Adj. R-squared	0.081	0.200	0.045

Table 5. The effects of firm/country characteristics: Public versus private firms.

This table presents the results of subsample analysis on two groups: publicly listed versus private firms. In columns 1 (4), 2 (5), and 3 (6), the dependent variables are Closure t, Sales growth t, and Employment growth t, respectively. The sources and detailed variable definitions are shown in the Appendix. Robust t-statistics based on standard errors clustered at the countryindustry level are reported in brackets. *, **, and *** indicate significance at the 10%, 5%, and

1% levels, respectively. All countries in our sample have both public and private firms.

1% levels, respect		Public Firm			Private Fir	ms
	Closure t	Sales growth t (2)	Employment growth t (3)	Closure t (4)	Sales growth t (5)	Employment growth t (6)
Ln (firm size) _{t-1}	-0.033***	3.167***	-3.073***	-0.026***	3.568***	-4.084***
	(-3.19)	(3.07)	(-3.44)	(-7.18)	(8.36)	(-6.91)
Ln (firm age) _{t-1}	-0.035*** (-2.64)	-3.046* (-1.66)	-0.730 (-0.54)	-0.018** (-2.29)	-0.575 (-0.79)	1.384* (1.93)
State t-1	-0.009 (-0.29)	12.583*** (3.28)	4.088 (1.26)	0.006 (0.17)	6.021* (1.93)	3.452 (0.87)
Foreign t-1	-0.017 (-0.67)	1.417 (0.54)	-2.556 (-0.86)	0.025** (2.58)	-0.830 (-0.48)	1.758* (1.81)
Subsidiary t-1	-0.015 (-0.67)	7.234*** (2.71)	9.193*** (4.33)	-0.035*** (-3.49)	3.191** (2.41)	4.981*** (4.50)
Exporter t-1	0.015 (0.55)	-2.904 (-1.04)	3.917 (1.29)	0.012 (1.24)	-1.267 (-1.25)	0.578 (0.59)
Finance obstacle t-1	0.007	-0.866	-0.588	0.011***	-1.197***	-0.251
COVID policy stringency t COVID spread t	(0.93) 0.002** (2.59) -0.001	(-0.98) -0.248*** (-3.66) -0.346	(-0.59) 0.025 (0.53) 0.377	(3.38) 0.002*** (5.35) -0.000	(-3.96) -0.314*** (-7.55) 0.288	(-0.88) -0.001 (-0.03) -0.575*
•	(-0.31)	(-0.54)	(0.62)	(-0.09)	(0.70)	(-1.94)
Industry FE	YES	YES	YES	YES	YES	YES
Macro controls	YES	YES	YES	YES	YES	YES
Observations	910	629	870	12171	8433	11453
Adj. R-squared	0.10	0.20	0.04	0.08	0.20	0.05

Table 6. The effects of firm/country characteristics: Pandemic vs. normal times.

This table presents the effects of firm/country characteristics on firm growth, side-by-side, in pre-pandemic (normal) and pandemic times. Columns (1) and (3) show results for the pandemic times, which are the same as columns (2) and (3) in Table 4. We report them here for ease of comparison. Columns (2) and (4) show results for the normal times. The sources and detailed variable definitions are shown in the Appendix. Robust t-statistics based on standard errors clustered at the country-industry level are reported in brackets. *, **, and *** indicate significance at the 10%, 5%, and 1% levels, respectively.

		ales growth		ployment growth
	Pandemic	Normal	Pandemic	Normal
	(1)	(2)	(3)	(4)
Ln (firm size) _{t-1}	3.423***	1.090***	-4.014***	1.186***
	(8.20)	(4.71)	(-7.25)	(7.83)
Ln (firm age) _{t-1}	-0.799	-4.569***	1.204*	-3.502***
	(-1.11)	(-11.29)	(1.68)	(-14.45)
State t-1	7.816***	2.357	5.918***	-1.922**
	(3.76)	(1.08)	(2.96)	(-2.06)
Foreign t-1	-0.262	0.275	1.158	-0.819**
	(-0.18)	(0.42)	(1.21)	(-2.35)
Subsidiary t-1	3.706***	-0.910	5.482***	-0.084
	(2.94)	(-1.51)	(5.46)	(-0.21)
Public t-1	-0.012	-1.642**	2.054	-0.570
	(-0.01)	(-2.23)	(1.63)	(-1.26)
Exporter t-1	-1.441	0.747	0.743	0.289
	(-1.53)	(1.48)	(0.79)	(1.01)
Finance obstacle t-1	-1.194***	-0.818***	-0.255	-0.512***
	(-4.13)	(-3.91)	(-0.92)	(-4.72)
COVID policy	0.040444			
stringency t	-0.310***		-0.003	
GOLIED 1	(-7.56)		(-0.08)	
COVID spread t	0.167		-0.525*	
	(0.44)		(-1.71)	
Ln (GDP per capita) _{t-1}	6.753***	0.910	4.298**	0.897*
	(4.04)	(0.87)	(2.37)	(1.71)
Age65 _{t-1}	0.123	-0.106	-0.165	-0.119*
	(0.46)	(-0.59)	(-0.56)	(-1.72)
Industry FE	YES	YES	YES	YES
Observations	9062	14387	12323	16476
Adj. R-squared	0.20	0.03	0.05	0.04

Table 7. The joint effects of COVID policy stringency and firm characteristics

This table presents the joint effects of government *COVID policy stringency* and firm characteristics on firm survival and growth in pandemic times. Columns (1), (2), and (3) reports results for dependent variables *Closure*, *Sales growth*, and *Employment growth*, respectively. The key independent variable is *COVID policy stringency*, a composite measure taking a value from 0 to 100 (100 = strictest), and its interactions with a set of firm characteristics. The sources and detailed variable definitions are shown in the Appendix. Robust t-statistics based on standard errors clustered at the country-industry level are reported in brackets. *, **, and *** indicate significance at the 10%, 5%, and 1% levels, respectively.

	Closure t	Sales growth t	Employment growth t
	(1)	(2)	(3)
COVID policy stringency t	0.003***	-0.430***	-0.089
	(2.69)	(-4.05)	(-0.77)
COVID spread t	-0.000	0.154	-0.548*
	(-0.07)	(0.41)	(-1.88)
Ln (firm size) _{t-1} × COVID policy stringency $_t$	-0.001***	0.033*	-0.032
	(-3.80)	(1.77)	(-1.21)
Ln (firm age) _{t-1} × COVID policy stringency $_t$	0.000	0.001	0.062*
	(0.83)	(0.04)	(1.87)
State $_{t-1} \times \text{COVID}$ policy stringency $_t$	0.000	-0.188**	-0.060
	(0.07)	(-2.41)	(-0.73)
Foreign $_{t-1} \times COVID$ policy stringency $_t$	0.000	-0.051	0.008
	(0.15)	(-0.84)	(0.18)
Subsidiary $t-1 \times COVID$ policy stringency t	-0.001	-0.027	0.143***
	(-1.35)	(-0.41)	(3.25)
Public $t-1 \times COVID$ policy stringency t	-0.000	0.140**	0.006
	(-0.31)	(2.25)	(0.12)
Exporter $_{t-1} \times COVID$ policy stringency $_t$	0.001	-0.000	-0.057
	(1.57)	(-0.00)	(-1.49)
Finance obstacle $_{t-1} \times COVID$ policy stringency $_t$	-0.000	0.011	0.007
	(-1.43)	(0.62)	(0.53)
Ln (firm size) _{t-1}	0.006	1.572	-2.212
,	(0.82)	(1.51)	(-1.53)
Ln (firm age) _{t-1}	-0.033**	-0.853	-2.362
	(-2.15)	(-0.49)	(-1.60)
State t-1	-0.025	18.782***	8.070**
	(-0.79)	(4.18)	(2.25)
Foreign t-1	0.015	2.462	0.659
<i>5</i> · · ·	(0.72)	(0.72)	(0.28)
Subsidiary t-1	-0.000	5.357	-2.731
y v.	(-0.01)	(1.25)	(-1.03)
Public t-1	0.012	-7.877**	1.691
• •	(0.38)	(-2.25)	(0.60)
Exporter t-1	-0.018	-1.377	3.618*
1	(-1.06)	(-0.43)	(1.84)
Finance obstacle t-1	0.021**	-1.783*	-0.636
	(2.43)	(-1.80)	(-0.86)
	()	(/	()
Industry FE	YES	YES	YES
Macro controls	YES	YES	YES
Observations			
Observations Adj. R-squared	13081 0.08	9062 0.20	12323 0.05

Table 8. The effects of culture on firm survival/growth in pandemic times.

This table presents the effects of three culture dimensions, i.e., *Individualism* _{t-1}, *Uncertainty* avoidance _{t-1}, *Long-term orientation* _{t-1}, on firm survival and growth in pandemic times. Columns (1), (2), and (3) report results for dependent variables *Closure* _t, *Sales growth* _t, and *Employment growth* _t, respectively. For the sake of brevity, only the coefficients of culture variables, *COVID policy stringency* _t, and *COVID spread* _t are reported. Firm characteristics, macro-economic variables, and industry fixed effects are included in all regressions. The sources and detailed variable definitions are shown in the Appendix. Robust t-statistics based on standard errors clustered at the country-industry level are reported in brackets. *, **, and *** indicate significance at the 10%, 5%, and 1% levels, respectively.

	Closure t (1)	Sales growth t (2)	Employment growth t (3)
Individualism _{t-1}	0.000	0.074	-0.051
	(0.86)	(1.17)	(-1.02)
Uncertainty avoidance t-1	-0.000	0.145	0.101
	(-0.96)	(1.39)	(1.51)
Long-term orientation t-1	-0.001***	0.115***	0.070***
	(-4.58)	(2.65)	(2.93)
COVID policy stringency t	0.002***	-0.399***	-0.044
	(4.14)	(-5.75)	(-1.04)
COVID spread t	0.000	-0.380	-0.554**
	(0.14)	(-0.60)	(-2.53)
Firm controls	YES	YES	YES
Industry FE	YES	YES	YES
Macro controls	YES	YES	YES
Observations	11293	7555	10646
Adj. R-squared	0.08	0.22	0.05

Table 9. The effects of country governance on firm survival/growth in pandemic times.

This table presents the effects of country governance quality on firm survival and growth in the pandemic times. Columns (1), (2), and (3) report results for dependent variables *Closure t, Sales growth t*, and *Employment growth t*, respectively. Panels A-G shows the results for different measures of country governance: A for *Control of corruption*; B for *Government effectiveness*; C for *Political stability*; D for *Regulatory quality*; E for *Rule of law*; F for *Voice and accountability*; and G for the *Average WGI index*. For the sake of brevity, only the coefficients of country governance variables are reported. Firm characteristics, country characteristics, and industry fixed effects are included in all regressions. The sources and detailed variable definitions are shown in the Appendix. Robust t-statistics based on standard errors clustered at the country-industry level are reported in brackets. *, **, and *** indicate significance at the 10%, 5%, and 1% levels, respectively.

	Closure t	Sales growth t	Employment growth t
	(1)	(2)	(3)
	Panel A: (Control of corrupt	ion
Control of corruption t-1	0.037*	-4.500***	-4.918**
-	(1.95)	(-2.65)	(-2.46)
	Panel B: G	overnment effecti	veness
Government effectiveness t-1	0.019	-2.497	-6.047**
	(0.64)	(-0.97)	(-2.02)
	Panel C: P	olitical stability	
Political stability t-1	0.023	-0.501	2.367
·	(1.39)	(-0.19)	(1.34)
	Panel D: R	egulatory quality	
Regulatory quality t-1	0.046**	-4.901***	-7.249***
	(2.27)	(-2.95)	(-4.08)
	Panel E: R	ule of law	
Rule of law t-1	0.024*	-1.754	-4.774**
	(1.70)	(-0.96)	(-2.59)
	Panel F: V	oice and accounta	bility
Voice and accountability t-1	0.027***	-3.211***	-2.719***
·	(3.38)	(-2.72)	(-3.01)
	Panel G: A	verage WGI inde	X
Average WGI index t-1	0.046**	-4.491**	-5.928***
-	(2.49)	(-2.24)	(-2.61)
Firm controls	YES	YES	YES
Industry FE	YES	YES	YES
Country controls	YES	YES	YES

Appendix. Variable Definitions and Data Sources.

Variable	Definition	Source
Closure t	A. Closure and Growth Dummy variable that takes on the value 1 if this firm is currently temporarily closed, or permanently closed, 0 otherwise.	WBES(COVb0)
Sales growth t	The percentage of the sales increase, negative value means the sales decrease.	WBES(COVb2a, COVb2b, & COVb2c)
Employment growth t	$(Ln(Employee_{2020}+1)-Ln(Employee_{t-1}+1))/(2020-survey\ year+1)*100\%$	WBES(COVd1 and 11)
	B. Firm-level Characteristics	
Ln (firm size) _{t-1}	$Ln(Employee_{t-1})$	WBES(11)
Ln (firm age) _{t-1}	Ln(survey year–firm founding year+1)	WBES(b5)
State t-1	Dummy variable that takes on the value 1 if the firm is at least partially owned by government or state, 0 otherwise.	WBES(b2c)
Foreign t-1	Dummy variable that takes on the value 1 if the firm is at least partially owned by private foreign individuals, companies or organizations, 0 otherwise.	WBES(b2b)
Subsidiary _{t-1} Public _{t-1}	Dummy variable that takes on the value 1 if the firm is part of a large firm, 0 otherwise. Legal status of the firm is shareholding company with traded shares.	WBES(a7) WBES(b1)
Exporter t-1	Dummy variable that takes on the value 1 if the firm exports, 0 otherwise.	WBES(d3a)
Finance obstacle t-1	Categorical variable, used to measure "how much of an obstacle: access to finance?"	WBES(k30)
	C. Country Characteristics	
COVID policy stringency t	A composite measure based on nine response indicators including school closures, workplace closures, and travel bans, rescaled to a value from 0 to $100 (100 = \text{strictest})$.	Oxford COVID-19 Government Response Tracker Center for Systems
COVID spread t	Cumulative confirmed COVID-19 cases per thousand persons.	Science and Engineering at Johns Hopkins University
Ln (GDP per capita) t-1	The logarithm GDP of per capita (constant 2010 US\$).	WDI
Age65 t-1	Percentage of population aged above 65 among the total population of an economy.	WDI

Variable	Definition	Source
Individualism t-1	National Culture Indicator of the degree to which individuals are supposed to look after themselves or remain integrated into groups, usually around the family.	Hofstede Insights
Uncertainty avoidance t-1	National culture indicator that expresses the degree to which the members of a society feel uncomfortable with uncertainty and ambiguity.	Hofstede Insights
Long-term orientation t-1	National culture indicator that expresses how every society has to maintain some links with its own past while dealing with the challenges of the present and future.	Hofstede Insights
Control of corruption t-1	Control of corruption captures perceptions of the extent to which public power is exercised for private gain, including both petty and grand forms of corruption, as well as "capture" of the state by elites and private interests.	World Governance Indicator
Government effectiveness t-1	Government effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from political pressures, the quality of policy formulation and implementation, and the credibility of the government's commitment to such policies.	World Governance Indicator
Political stability t-1	Political stability measures perceptions of the likelihood of political instability and/or politically-motivated violence, including terrorism.	World Governance Indicator
Regulatory quality t-1	Regulatory quality captures perceptions of the ability of the government to formulate and implement sound policies and regulations that permit and promote private sector development.	World Governance Indicator
Rule of law t-1	Rule of law captures perceptions of the extent to which agents have confidence in and abide by the rules of society, and in particular the quality of contract enforcement, property rights, the police, and the courts, as well as the likelihood of crime and violence.	World Governance Indicator
Voice and accountability t-1	Voice and accountability captures perceptions of the extent to which a country's citizens are able to participate in selecting their government, as well as freedom of expression, freedom of association, and a free media.	World Governance Indicator
Average WGI index t-1	A proxy of a country's overall governance quality, defined as the average of the six country governance indexes.	Authors' own construct